Author's response to reviews

Title: How are researchers handling missing data in RCTs? A review of the top medical journals

Authors:

Melanie L Bell (melaniebell@email.arizona.edu)
Mallorie Fiero (mfiero@email.arizona.edu)
Nicholas J Horton (nhorton@amherst.edu)
Chiu-Hsieh Hsu (pchhsu@email.arizona.edu)

Version: 5 Date: 16 October 2014

Author's response to reviews:

Dear Editor

Thank you for the opportunity to revise our manuscript. As suggested by Arlene Pura we have removed the "funding" and "ethics" section at the end of the manuscript. We have not included an acknowledgement section as it was not appropriate for this paper.

Please let me know if there is anything else you need.

Best regards

Melanie Bell